Author's response to reviews

Title: Capsular Synovial Metaplasia Mimicking the Silicone Leak of a Breast Prosthesis

Authors:

Sarah Krishnanandan (sarah.abbassian@gmail.com)
Ali Abbassian (aabbassian@gmail.com)
Anup K Sharma (noopysharma@hotmail.com)
Giles Cunnick (gcunnick@doctors.org.uk)

Version: 3 Date: 18 June 2008

Author's response to reviews:

Revision was undertaken as suggested

Ref 1: No specific suggestions
Ref 2: Extra paragraph added to conclusions to explain that when metaplasia is confirmed explantation should be delayed and the patient observed awaiting resolution of the leak.

Advice was given by a histopathologist in the preparation of the slides.